Author's response to reviews

Title: Platelet-activating factor receptor (PAF-R)-dependent pathways control tumour growth and tumour response to chemotherapy

Authors:

Soraya I Oliveira (sorayaio@if.usp.br)
Luciana NS Andrade (lnsa@usp.br)
Ana C Onuchic (ana_onuchic@hotmail.com)
Suely Nonogaki (nonononogaki@terra.com.br)
Patricia D Fernandes (cpattern@farmaco.ufrj.br)
Monica C Pinheiro (rchammas@lim24.fm.usp.br)
Ciro BS Rohde (cirobsr@gmail.com)
Roger Chammas (rchammas@lim24.fm.usp.br)
Sonia Jancar (sojancar@icb.usp.br)

Version: 6 Date: 23 March 2010

Author's response to reviews: see over
Dear Editor,

We were very happy to hear that our manuscript was accepted for publication in BMC Cancer after minor revision. The revision asked by reviewers 1 and 3 was made and changes are highlighted with yellow marker. Citations made in the text as numbers were put in brackets instead of parenthesis. For language correction we hired the Academic Editing Service, UK, and their corrections are marked in red. The Authors’ Contribution section was re-written to conform to the format of the journal.

We hope that the manuscript can be now formally accepted and look forward your reply.

Sincerely yours

Prof. Dr. Sonia Jancar